

## Portopulmonary hypertension in the current era of pulmonary hypertension management

Laurent Savale, Manuel Guimas, Nathan Ebstein, Marie Fertin, Mitja Jevnikar, Sébastien Renard, Delphine Horeau-Langlard, Cécile Tromeur, Céline Chabanne, Grégoire Prevot, et al.

#### ▶ To cite this version:

Laurent Savale, Manuel Guimas, Nathan Ebstein, Marie Fertin, Mitja Jevnikar, et al.. Portopulmonary hypertension in the current era of pulmonary hypertension management. Journal of Hepatology, 2020, 73 (1), pp.130-139. 10.1016/j.jhep.2020.02.021. hal-04277691

### HAL Id: hal-04277691 https://hal.science/hal-04277691v1

Submitted on 29 Nov 2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial 4.0 International License

1

#### **Portopulmonary hypertension in the current era of pulmonary**

#### hypertension management

Laurent Savale<sup>1,2,3</sup>, Manuel Guimas<sup>4#</sup>, Nathan Ebstein<sup>1,2 #</sup>, Marie Fertin<sup>5</sup>, Mitja Jevnikar<sup>1,2,3</sup>,

Sébastien Renard<sup>6</sup>, Delphine Horeau-Langlard<sup>7</sup>, Cécile Tromeur<sup>8</sup>, Céline Chabanne<sup>9</sup>,

Grégoire Prevot<sup>10</sup>, Ari Chaouat<sup>11</sup>, Pamela Moceri<sup>12</sup>, Élise Artaud-Macari<sup>13</sup>, Bruno Degano<sup>14</sup>,

Romain Tresorier<sup>15</sup>, Clément Boissin<sup>16</sup>, Hélène Bouvaist<sup>17</sup>, Anne-Claire Simon<sup>18</sup>,

Marianne Riou<sup>19</sup>, Nicolas Favrolt<sup>20</sup>, Sylvain Palat<sup>21</sup>, Delphine Bourlier<sup>22</sup>, Pascal Magro<sup>23</sup>,

Vincent Cottin<sup>24</sup>, Emmanuel Bergot<sup>4</sup>, Nicolas Lamblin<sup>5</sup>, Xavier Jaïs<sup>1,2,3</sup>, Audrey Coilly<sup>25</sup>, François Durand<sup>26</sup>, Claire Francoz<sup>26</sup>, Filomena Conti<sup>27</sup>, Philippe Hervé<sup>1,2,3</sup>, Gérald Simonneau<sup>1,2,3</sup>, David Montani<sup>1,2,3</sup>, Jean-Charles Duclos-Vallée<sup>25</sup>, Didier Samuel<sup>25</sup>, Marc Humbert<sup>1,2,3</sup>, Pascal De Groote<sup>5</sup>, Olivier Sitbon<sup>1,2,3</sup>

# These authors have contributed equally to this work.

- <sup>1</sup> Université Paris-Saclay, Faculté de Médecine, Le Kremlin Bicêtre, France.
- <sup>2</sup> AP-HP, Service de pneumologie, Centre de Référence de l'Hypertension Pulmonaire, Hôpital Bicêtre, Le Kremlin Bicêtre, France.
- <sup>3</sup> INSERM UMR\_S 999, Hôpital Marie Lannelongue, Le Plessis Robinson, France.
- <sup>4</sup> Service de Pneumologie, CHRU Côte de Nacre, 14033 Caen, France.
- <sup>5</sup> Université de Lille, Service de cardiologie, CHU Lille, Institut Pasteur de Lille, Inserm U1167, F-59000, Lille, France.
- <sup>6</sup> Département de cardiologie, Hôpital La Timone, Aix-Marseille Université, Marseille, France.
- <sup>7</sup> Service de pneumologie, CHU de Nantes, Hôpital Laënnec, Nantes, France.
- <sup>8</sup> European Brittany University, Brest, France; Department of Internal Medicine and Chest Diseases, University Hospital Centre La Cavale Blanche, Brest, France; Groupe d'Etude de la Thrombose de Bretagne Occidentale (GETBO), EA 3878, CIC INSERM 1412, Brest, France.
- <sup>9</sup> Département de cardiologie et maladies vasculaires, CHU de Rennes Hôpital Pontchaillou, Rennes, France.
- <sup>10</sup> CHU de Toulouse, Hôpital Larrey, Service de pneumologie, Toulouse, France.
- <sup>11</sup> Université de Lorraine; Département de pneumologie, CHRU de Nancy; Inserm U1116, Vandœuvre-lès-Nancy, France.
- <sup>12</sup> Service de cardiologie, Centre Hospitalier Universitaire de Nice-Hôpital Pasteur, Nice, France.
- <sup>13</sup> Département de pneumologie, Hôpital universitaire, Rouen, France.
- <sup>14</sup> Service de pneumologie, CHRU, Besançon, France.
- <sup>15</sup> Service de cardiologie, Hôpital Gabriel Montpied, CHU, Clermont-Ferrand, France.

- <sup>16</sup> Service de pneumologie, Hôpital universitaire, Montpellier, France.
- <sup>17</sup> Hôpital La Tronche, Service de cardiologie, CHU de Grenoble-Alpes, Grenoble, France.
- <sup>18</sup> Service de pneumologie, CHU Poitiers, Poitiers, France.
- <sup>19</sup> Service de pneumologie, nouvel hôpital civil, hôpitaux universitaires, Strasbourg, France.
- <sup>20</sup> Service de pneumologie, CHU, Dijon, France.
- <sup>21</sup> Service de pneumologie, CHRU, Limoges, France.
- <sup>22</sup> Service des maladies respiratoires, hôpital Haut-Lévêque, CHU de Bordeaux, Pessac, France.
- <sup>23</sup> Service de pneumologie, CHRU de Tours, Tours, France.
- <sup>24</sup> Université Claude-Bernard Lyon 1, hôpital Louis-Pradel, service de pneumologie, Centre de référence national des maladies pulmonaires rares, UMR154, 69677 Lyon, France.
- <sup>25</sup> AP-HP Hôpital Paul-Brousse, Centre Hépato-Biliaire, Villejuif; UMR-S 1193, Université Paris-Saclay; DHU HEPATINOV, Villejuif, France.
- <sup>26</sup> Inserm, U1149, Centre de Recherche sur l'Inflammation (CRI) UMRS1149, Université de Paris, AP-HP, Service d'hépatologie, Hôpital Beaujon, Clichy, France
- <sup>27</sup> Sorbonne Université, UPMC, Inserm, UMR\_S 938 "Centre de Recherche Saint-Antoine", Paris, France; AP-HP, Pitié-Salpêtrière Hospital, Unité Médicale de Transplantation Hépatique, Paris, France.

#### Correspondence and reprint request:

Dr Laurent Savale

Service de Pneumologie et Soins Intensifs Respiratoires, Hôpital Bicêtre,

78, Rue du Général Leclerc, 94270 Le Kremlin-Bicêtre, France.

Tel: +33.1.45.21.79.72; Fax: +33.1.45.21.79.71; Email: <u>laurent.savale@aphp.fr</u>

Key words: Hypertension, pulmonary - Pulmonary arterial hypertension - Portopulmonary

 $hypertension-Survival-Prognostic\ factors-Cirrhosis-Liver\ transplantation$ 

Text Word Count: 5 542 words

Abstract Word Count: 267 words

Number of figures and tables: 8

**Conflict of interest statement:** 

LS reports grants and personal fees from Actelion, grants and personal fees from MSD, grants and personal fees from GSK, outside the submitted work.

MH reports grants and personal fees from Actelion, grants and personal fees from Bayer, grants from GSK, grants and personal fees from MSD, personal fees from Ferrer, personal fees from Acceleron, outside the submitted work.

GS reports grants and personal fees from Actelion, grants and personal fees from Bayer, grants from GSK, grants and personal fees from MSD, outside the submitted work. OS reports grants and personal fees from Actelion, grants and personal fees from Bayer, grants from GSK, grants and personal fees from MSD, personal fees from Ferrer, personal fees from Gossamer Bio, personal fees from Acceleron, outside the submitted work. DM reports grants and personal fees from Actelion, grants and personal fees from Bayer, grants and personal fees from GSK, personal fees from MSD, outside the submitted work. PDG reports personal fees from Actelion, MSD, Bayer, outside the submitted work. XJ reports grants and personal fees from Actelion, grants and personal fees from Bayer, grants and personal fees from GSK, personal fees from Merck, outside the submitted work. DS reports personal fees from Intercept, personal fees from Biotest, personal fees from Abbvie, personal fees from Gilead sciences, outside the submitted work. PDG reports personal fees and non-financial support from ACTELION, personal fees and non-financial support from BAYER - MSD, personal fees and non-financial support from NOVARTIS, personal fees and non-financial support from VIFOR, personal fees from ABBOTT, personal fees and non-financial support from SERVIER, personal fees from ASTRA-ZENECA, non-financial support from AMGEN, outside the submitted work.

NL reports personal fees and other from MSD, personal fees from AMICUS

THERAPEUTICS, other from BMS, personal fees from NOVARTIS, personal fees from

ASTRA-ZENECA, personal fees from BAYER, personal fees from AKCEA, outside the submitted work.

MG, NE, MF, MJ, SR, DHL, CT, CC, GP, AC, PM, EAM, BD, RT, CB, HB, ACS, MR, NF, SP, DB, PM, VC, EB, AC, FD, CF, PH, JCDV have nothing to disclose

Funding Sources: None

**Authors' contributions**: LS and OS were responsible for design, analysis and interpretation of data. All authors participated to data collection. LS and OS prepared the manuscript. All authors were responsible for manuscript review and revision. All authors approved the final version of the manuscript.

#### ABSTRACT

**Background & Aims.** Long-term outcomes in portopulmonary hypertension (PoPH) are poorly studied in the current management era of pulmonary hypertension. We analysed the effect of pulmonary arterial hypertension (PAH)-targeted therapies, the survival and the predictors of death in a large contemporary cohort of patients with PoPH.

**Methods.** Data from patients with PoPH consecutively enrolled in the French Pulmonary Hypertension Registry between 2007 and 2017 were collected. Effect of initial treatment strategies on functional class, exercise capacity and cardiopulmonary haemodynamics were analysed. Survival and its association with PAH- and hepatic-related characteristics were also examined.

**Results.** Six hundred and thirty-seven patients (mean age  $55\pm10$  years; 58% male) were included. Fifty seven percent had mild cirrhosis, i.e. Child-Pugh (CP) stage A. The median MELD score was 11 (Q1-Q3; 9-15). Most patients (n=474; 74%) were initiated on monotherapy, either with a phosphodiesterase-5 inhibitor (n=336) or with an endothelin-receptor antagonist (n=128); 95 (15%) were initiated on double oral combination therapy and 5 (1%) on triple therapy. After a median treatment time of 4.5 months, there were significant improvements in functional class (p<0.001), 6-minute walk distance (6MWD) (p<0.0001) and pulmonary vascular resistance (p<0.0001). Overall survival rates were 84%, 69% and 51% at 1, 3 and 5 years, respectively. Baseline 6MWD, sex, age and MELD Score or CP stage were identified as independent prognostic factors. Survival from PoPH diagnosis was significantly better in the subgroup of patients who underwent liver transplantation (92%, 83% and 81% at 1, 3 and 5 years, respectively)

**Conclusion.** Survival of patients with PoPH is strongly associated with the severity of liver disease. Patients who underwent liver transplantation had the best long-term outcomes.

#### Lay Summary

Portopulmonary hypertension (PoPH) is defined by the presence of pulmonary arterial hypertension (PAH) in a context of chronic liver disease, and is characterized by progressive shortness of breath and exercise limitation. The presence of severe PAH in liver transplant candidates represents a contraindication for such a surgery. PAH-targeted medications are notably efficacious in PoPH allowing liver transplant candidates to undergo safe transplantation. The combination of PAH medications followed by liver transplantation are associated with a good survival in PoPH

#### **INTRODUCTION**

Portopulmonary hypertension (PoPH) is the concomitant presence of pulmonary arterial hypertension (PAH) and portal hypertension in patients with or without cirrhosis. PAH can develop in 2 to 6% of patients with portal hypertension and results from complex pathophysiological interactions between the portal and pulmonary circulation (1). Based on the high prevalence of cirrhosis, it is likely that PoPH remains underdiagnosed in many countries (2). In pulmonary hypertension (PH) registries, PoPH patients represent 5 to 15% of PAH patients (3–5). Survival and prognostic factors in PoPH remain controversial and are still poorly studied in the current era of PH management (3–5). Insights from previous PoPH registries are limited by the small number of patients enrolled, the historical nature of the data (ie, not from the current treatment era) and by the lack of information on the severity of the underlying disease (3–5).

The management of patients with PoPH is complex and must be based on a multidisciplinary approach, especially for liver transplant candidates (6). Although evidence supporting the use of PAH-targeted therapies in patients with PoPH is limited, the last joint European Society of Cardiology / European Respiratory Society (ESC/ERS) guidelines state that the algorithm for the management of PAH patients can be cautiously applied to patients with PoPH, taking into account the severity of the liver disease (7,8). The evidence for the use of PAH-targeted therapies in PoPH are scarce and mainly comes from retrospective experiences/analyses in/of small patient/case series. Recently, a dedicated randomized controlled trial demonstrated that, compared to placebo, a 12-week treatment with macitentan, a dual endothelin-receptor antagonist (ERA), significantly lowered pulmonary vascular resistance, without hepatic safety concerns (9). Finally, there is few information on the long-term use of PAH-targeted therapies in patients with PoPH.

With the development of efficacious PAH-targeted medications, the prognosis of patients with PoPH may have changed over the last decades. The current study was designed to evaluate the clinical and hemodynamic effect of PAH-targeted therapies and survival in a large contemporary cohort of patients with PoPH.

#### **PATIENTS AND METHODS**

#### **Study population**

The study population consisted in incident, treatment-naïve patients with PoPH who were prospectively enrolled in the French PH Network registry between January 2007 and January 2017 (Supplementary CTAT Table). Eligible patients had newly diagnosed precapillary PH associated with portal hypertension with or without cirrhosis, or congenital portosystemic shunts. Precapillary PH was defined as a resting mean pulmonary artery pressure (mPAP)  $\geq$ 25 mmHg with pulmonary artery wedge pressure (PAWP)  $\leq$ 15 mmHg and pulmonary vascular resistance (PVR) >3 Wood units, measured by right heart catheterization (RHC) (7). The diagnosis of portal hypertension was based on hemodynamic measurement of a hepatic venous pressure gradient of more than 5 mmHg, or the combination of suggestive signs, including the presence of splenomegaly, thrombocytopenia and/or oesophageal varices, or clinical signs of portosystemic shunt. Cirrhosis was diagnosed by a documented historical liver biopsy or the presence of typical clinical and/or biological characteristics. Patients with extrahepatic portal hypertension (like portal vein thrombosis) or congenital portosystemic shunts who developed PAH were also classified as PoPH and eligible.

In line with the current guidelines, all patients with PH confirmed by RHC underwent extensive investigation to identify other causes or risk factors. Patients classified as group 2 (secondary to left heart disease), group 3 (due to lung disease or chronic hypoxemia), group 4

(chronic thromboembolic pulmonary hypertension) PH or group 5 (PH with unclear and/or multifactorial mechanisms), or with missing hemodynamic variables at enrollment in the PH registry were excluded from the study.

This study complied with the Declaration of Helsinki. Although French law does not require ethics committee approval or informed consent for retrospective data collection, the data collected were anonymized and complied with the requirements of the *'Commission Nationale Informatique et Libertés'*, the organization dedicated to privacy, information technology and civil rights in France. The committee approved the methods used to collect and analyze the data on 24 May 2003 (approval number 842063).

#### Variables collected

Variables related to liver disease were assessed at time of PoPH diagnosis. Aetiology of portal hypertension was systematically reported. Severity of cirrhosis was assessed by Child-Pugh stage and/or Model for End-Stage Liver Disease (MELD) score which were calculated post-hoc based on the relevant available information. Liver transplant candidates were identified, and the date of transplantation was recorded if performed.

Variables related to PAH were assessed at baseline and at the first follow-up visit within 12 months of diagnosis. Modified New York Heart Association (NYHA) functional class, physical examination, routine blood tests, non-encouraged 6-min walk test and standard cardiopulmonary hemodynamic variables assessed by RHC were recorded.

#### **Evaluation of initial PAH therapy**

Patients were categorized according to the type and number of PAH-targeted therapy initiated within 6 months of diagnostic RHC: 1) monotherapy with an ERA, a phosphodiesterase type-5 inhibitor (PDE-5i), or a prostacyclin analogue; 2) double or 3) triple combination therapy.

Effect of the different initial treatment strategies on clinical status, exercise capacity and haemodynamics were analysed at the first follow-up visit within 12 months of diagnosis.

#### Statistical analysis

Continuous variables were expressed as mean ± SD for normally distributed variables or median (first and third interquartile) for non-normally distributed variables. Categorical variables were expressed as number of patients and relative frequencies (percent). Differences in continuous variables were compared using the independent Student's t-test for normally distributed variables and the Mann–Whitney U test for non-normally distributed variables. Changes from baseline to follow-up were assessed using a paired Student's t test, Wilcoxon signed-rank test, or Chi-square test as appropriate. All comparisons were two sided with a pvalue <0.05 considered significant.

Overall survival time was defined from the date of diagnostic RHC to death. Surviving patients were censored at the date of last clinical contact. Patients who underwent liver transplantation (LT) were not censored. The Kaplan-Meier method was used to estimate the proportion of patients surviving at each time point up to 5 years post-diagnosis. Survival curves were compared with log-rank test. Univariable analysis based on the Cox proportional hazards model was used to examine the relationship between survival and selected demographic, medical history and baseline variables. Multivariable analyses based on the Cox proportional hazards regression model were used to examine the independent effect of each variable on survival, controlling for possible confounding variables. Variables with a p-value  $\leq 0.1$  in the univariable analysis were eligible for entry into the multivariable models only if they were not highly correlated (absolute value of Pearson's or Spearman's *rho* <0.6) with other continuous variables and if <25% of individuals had missing values for that variable.

#### RESULTS

#### **Study population**

Patient disposition is shown in **Figure 1**. During the 10-year inclusion period, 4225 incident patients with PAH were enrolled in the French PH network registry. Of these, 742 patients had a diagnosis of portal hypertension (17.6%). After exclusion of patients with missing data or incompatible hemodynamic variables, study population comprised 637 patients with PoPH. Demographic and hepatic-related characteristics at PoPH diagnosis are shown in **Table 1**. The mean ( $\pm$ SD) age was 55 $\pm$ 10 years and the proportion of male was 58%. The majority of patients had alcohol cirrhosis. The proportion of patients with cirrhosis Child-Pugh stage B or C was 43% and that of patients with a MELD score  $\geq$ 15 was 30%. PAH-related characteristics are shown in **Table 1**. The median (Q1, Q3) baseline 6-minute walk distance (6MWD) was 355 m (252, 425), and 63.4% of patients presented in NYHA functional class III or IV.

#### Effect of initial treatment strategies

Initial treatment strategy is shown in **Figure 1**. Ninety percent of patients received PAHtargeted therapy within 6 months of PoPH diagnosis. The proportion of patients in Child-Pugh class C who were not treated was 16% (9/57) versus 9% in the group of patients in Child-Pugh class A (30/333) and Child-Pugh class B (17/197) (p=0.23). The most frequently used strategy was initial oral monotherapy (74.4 %), with a PDE-5i (sildenafil, n=239; tadalafil, n=97) or an ERA (bosentan, n=90; ambrisentan, n=36; macitentan, n=2). Double combination was initiated in 95 patients (bosentan and sildenafil, n=31; bosentan and tadalafil, n=15; ambrisentan and sildenafil, n=17; ambrisentan and tadalafil, n=24; macitentan and tadalafil, n=2). Upfront triple combination therapy was used for only 5 patients (0.8 %). Comparison of patient characteristics who received initial oral monotherapy with an ERA, initial oral

monotherapy with a PDE-5i and double combination with an ERA and a PDE-5i is shown in **Table S1**.

Repeated clinical assessment was performed after a median (Q1, Q3) period of 4.5 (3.2, 7.4) months in 504 out of the 574 patients initiated on PAH-targeted therapy. RHC at the time of first follow-up was performed for 453 patients. Seventy patients were not reassessed within one year from diagnosis due to death (n=40) or loss to follow-up (n=30).

At the time of first follow-up, 264 out of 504 patients (52%) improved symptoms by at least one functional class. The median change (Q1, Q3) in 6MWD from baseline to first assessment was +40 meters (-9, +101). A significant effect of treatment on all hemodynamic variables was observed as well as a significant increase in the proportion of patients with normal level of brain natriuretic peptide (BNP) or N-terminal pro-brain natriuretic peptide (NT-proBNP) (**Table 2**).

Changes in PVR, mPAP, cardiac output (CO) and 6MWD observed with an ERA, a PDE-5i or the combination of both are illustrated in **Figure 2**. Changes in PVR and 6MWD at first follow-up were similar in patients treated with an ERA and those treated with a PDE-5i. Despite a higher mean baseline PVR, patients treated with double oral combination had a significantly greater median change in PVR than patients initiated on monotherapy with either ERA (- 64% vs -40%, p<0.0001) or a PDE-5i (-64% vs -37%, p<0.0001).

The effect of ERA or PDE-5i on PVR was consistent across different subgroups including age, gender, aetiology of cirrhosis, MELD score, Child-Pugh class, functional class, 6MWD and associated HIV infection (**Figure S1**).

#### Survival and prognostic factors

Over a median (Q1, Q3) follow-up period of 35 (15, 63) months, 136 (21%) patients needed treatment escalation with PAH-targeted medications. Three hundred and two patients (47%)

died during the observation period. Overall survival rates were 84%, 69% and 51% at 1, 3 and 5 years, respectively (**Figure 3A**). The median survival time was 60.9 months. The univariable Cox regression analysis for the association of baseline characteristics and overall survival is shown in **Table 3**. In multivariable models, age, sex, 6MWD and MELD score (model A) or Child-Pugh stage (model B) were independently associated with survival. (**Table 3**). Overall survival according to the severity of the underlying liver disease is shown

#### in Figures 3B, 3C and 3D.

Causes of death were recorded in 184/302 cases (61%) and reported in **Table S2**. The main causes of death differed according to the severity of the underlying disease. Patients with mild cirrhosis mainly died from PAH or cancers (hepatocarcinoma and extrahepatic cancers), while patients with more severe cirrhosis mainly died from liver disease.

#### Survival of liver transplantation recipients

PoPH was diagnosed during a pretransplant assessment in 117 patients. Of these, 63 (54%) underwent LT during the follow-up period. Most of them (n=60, 95%) received PAH-targeted therapy (46 on monotherapy, 12 on double combination and 2 on triple combination therapy) as a bridge to LT. At the time of LT, 15 patients (24%) received intravenous epoprostenol that was initiated as first line therapy in 6 of them and as add-on therapy in the 9 remaining. Baseline characteristics of transplanted patients were compared to those of patients who had no indication for LT and those who had an indication for transplantation but who have not been transplanted (**Table S3**).

Overall survival of patients who underwent LT was 92%, 83% and 81% at 1, 3 and 5 years from PoPH diagnosis, respectively. In comparison, patients who had an indication for transplantation at time of PoPH diagnosis but did not undergo transplantation had a 1-, 3- and 5-year survival rate of 65%, 39% and 20%. Interestingly, survival from PoPH diagnosis was also better in transplanted patients when compared to than that of patients who had no indication for transplantation (92%, 83%, 81% vs 85%, 70%, 50% survival at 1 or 3 or 5 years, respectively, p<0.0001) (**Figure 4A**). Moreover, survival from PoPH diagnosis was better in all patients who underwent LT irrespective of Child-Pugh stage or MELD score when compared to patients who were not transplanted (**Figure 5**). Survival from LT was 88%, 85% and 78% at 1 or 3 or 5 years, respectively (**Figure 4B**). Twelve patients have died after LT including six deaths related to pulmonary arterial hypertension, all occurring within the first month following LT. Among survivors, PAH therapy was simplified from combination to monotherapy in 10 patients (16%), and discontinued in 14 (22%). Intravenous epoprostenol was weaned off after 1 to 23 months post-LT in the 12 survivors who received this treatment at the time of LT

#### DISCUSSION

This study describes the largest cohort of patients with PoPH to date. It provides a contemporary overview of long-term outcomes in patients with PoPH and identifies the main prognostic factors in the era of modern PAH management. In our study, the vast majority of patients received monotherapy with a PDE-5i or an ERA, both leading to similar clinical and hemodynamic improvement. Importantly, and as in other forms of PAH, double oral combination of these drugs led to a greater decrease in PVR than monotherapy. Survival was highly associated with the severity of the underlying liver disease. Other identified prognostic factors were age, gender and baseline 6MWD. The survival of patients who underwent LT was better than that of non-transplanted PoPH patients.

The proportion of PoPH patients in the French PH registry has progressively increased over the last decade, resulting in PoPH being today the third aetiology of incident PAH cases in the French PH registry. This is contrast with other registries and may be explained by the extensive collaboration between hepatologists and PAH expert centers in France, which is critical to improving detection and management of this condition (4,10–12). The differences in PoPH prevalence observed between countries may also be due to variability in health systems, screening approaches as well as in the prevalence of cirrhosis (1). Despite the absence of PoPH patients in the main randomized controlled trials for PAH therapies, about 90% of our patients received PAH-targeted therapy according to current ERS/ESC guidelines (7). The vast majority of our patients was initiated on monotherapy. This is in line with the recent recommendations from the 6<sup>th</sup> World Symposium on PH which underline a residual role for monotherapy in subsets of patients, including those with PoPH (13). Most of these patients received initial therapy with a PDE-5i; this choice may have been

driven by the absence of hepatotoxicity with this class of drug. In our study, a significant proportion (35%) of patients also received an ERA either as monotherapy or in combination with a PDE-5i or a prostacyclin analogue. We previously observed that the safety profile of bosentan in patients with PoPH and cirrhosis Child-Pugh stage A or B was generally consistent with that observed in PAH patients without cirrhosis (14). Interestingly, the effect of monotherapy with a PDE-5i on clinical parameters and cardiopulmonary hemodynamic was similar to that of monotherapy with an ERA. Recently, the first randomized placebo-controlled trial dedicated to patients with PoPH (PORTICO) demonstrated that a 12-week treatment with macitentan led to a significant 35% decrease in PVR, with no hepatic safety concerns (9). As macitentan is not currently available in France, our cohort included a very small number of patients treated with macitentan (ie, patients enrolled in the PORTICO study). Nevertheless, in our study, monotherapy with an ERA led to a change in PVR that was similar to that observed in PORTICO.

Our data also support the use of initial combination therapy in selected PoPH patients. This treatment strategy was mainly used in patients with high PVR. With a mean decrease in PVR of about 60%, double combination therapy led to a better hemodynamic improvement than monotherapies, as observed in other forms of PAH (15). Overall, specific considerations, including hemodynamic profile and severity of the cirrhosis, must be taken into account in the management of PoPH patients. Fluid retention and/or worsening of portal hypertension induced by vasodilatation in the splanchnic circulation may occur with vasodilators (16–18). In addition, combination therapies must be used with caution to avoid high cardiac output, as the long-term effect of this condition on the pulmonary circulation and heart function is unknown.

Use of PAH-targeted therapies was also associated with a significant improvement in exercise capacity and functional class, irrespective of initial treatment strategy. In contrast, the

PORTICO trial with macitentan did not show significant effect on functional class and walk distance (9). However, it should be noted that PORTICO enrolled a relatively small number of patients who were predominantly prevalent, with most patients receiving background therapy with a PDE-5i.

Previous studies based on PoPH registries reported different survival rates, with 5-year survival rates ranging from 35% to 68% (3–5). More than 10 years ago, Le Pavec et al. showed, in a French series of 154 PoPH patients, that survival depended on the severity of both the pulmonary hypertension and the underlying liver disease (3). In contrast, in the UK PH registry, no clear prognostic factors were identified in 110 PoPH patients with hemodynamic and liver disease severity comparable to that of the French PoPH cohort described by Le Pavec et al. (5). Finally, data from the US REVEAL registry showed a worse survival for 174 PoPH patients as compared to patients with idiopathic PAH (4). It is important to note that the lack of information on the severity of the underlying liver disease in the REVEAL registry limits the interpretation of these results. In addition, the large periods of inclusion and the frequent missing information on the severity of the underlying liver disease may explain this broad range of survival rates. In our study, the 5-year survival was 51% and one of the main prognostic factors was the severity of the underlying liver disease. The large number of patients included in our study and the extent of the information available on the severity of the underlying liver disease allowed a robust analysis of prognostic factors in the era of modern PAH management. Interestingly, hemodynamic variables assessed at PoPH diagnosis were not independently associated with the risk of death. This could be due to a more aggressive therapeutic approach in patients with more severe haemodynamics at baseline.

The weight of liver disease severity on prognosis underlines the need for a multidisciplinary approach in the management of PoPH patients, including PAH experts and hepatologists. In our study, the main causes of death were complications of liver disease, in particular in patients with advanced cirrhosis. In contrast, the main causes of death in patients with mild cirrhosis were pulmonary hypertension and extrahepatic cancers.

Before the availability of PAH specific therapies, LT was usually contraindicated in most patients with PoPH because the hemodynamic severity was associated with an unacceptable risk of postoperative right heart failure (19). The International Liver Transplant Society Practice Guidelines recently underlined that "unlike hepatopulmonary syndrome, there are no data to support the concept that PoPH treated or not treated should be an indication for LT" (6). In the era of modern PAH therapies, we observed that survival of PoPH patients who underwent LT was better than non-transplanted patients, including those with mild cirrhosis. This observation likely results from a more aggressive management of the PoPH with PAHtargeted therapies as a bridge to LT to improve cardiopulmonary haemodynamics and reduce the risk of perioperative right heart failure. It is interesting to underline that only 24% of transplanted patients were using intravenous epoprostenol as a bridge to LT. In the era of current PAH management, haemodynamic criteria allowing a safe transplantation can be reached with oral therapies in the majority of LT candidates. Moreover, the possibility to obtain a priority access to LT using an expert component ("MELD exception") in this specific indication likely contributed to reduce the mortality on waiting list and to improve the overall survival from PoPH diagnosis of the subgroup of patients who were candidates for LT. In addition, several studies have shown that stabilization, improvement or normalization of cardiopulmonary haemodynamics after LT is an attainable goal (20–23). The survival benefit

of LT in PoPH patients raises the question of whether LT should be considered in selected PoPH patients who do not have a liver transplant indication because of mild liver disease.

The main limitations of our study were its retrospective nature and the absence of collection of adverse events, such as hospitalization for PAH worsening, a major prognostic factor in PAH (24), and side effects related to therapies which are important to consider for the management of PoPH patients. We also want to underline that initial treatment strategy may explain the lack of independent association between baseline hemodynamics and outcomes, given that disease severity has likely influenced treatment decisions. Similar findings have been observed in idiopathic PAH or PAH associated with scleroderma (25–27). Finally, the better survival of transplanted patients may be driven by the selection of patients without severe comorbidities. Moreover, due to a systematic screening for PAH in liver transplant candidates, patients who were transplanted had less severe pulmonary hypertension at the time of PoPH diagnosis (**Table S3**).

In conclusion, our study confirms that PoPH is a devastating complication of portal hypertension, irrespective of the aetiology or the severity of the underlying liver disease. Our contemporary data from the largest cohort of PoPH patients to date confirm the effectiveness of PAH-targeted therapies on clinical, functional and hemodynamic variables without class effect. Prognostic factors are strongly related to the severity of cirrhosis. This underscores the need for a multidisciplinary approach including hepatologists and PAH experts for the management of PoPH patients. Finally, combination of PAH-targeted therapy and LT led to excellent long-term outcomes in selected PoPH patients.

#### Acknowledgements

The authors acknowledge Laurence Rottat for her help in obtaining the data for this study and her hard work in managing data in the French PH Registry. They also thank all participants to the French PH Network and liver transplant centers : Fabrice Bauer (Rouen), Laurent Bertoletti (Saint Etienne), Philippe Bonniaud (Dijon), Philippe Brosset (Limoges), Arnaud Bourdin (Montpellier), Yvon Calmus (CHU La Pitié salpêtrière), Matthieu Canuet (Strasbourg), Claire Dauphin (Clermont-Ferrand), Claire Dromer (Bordeaux), Christophe Duvoux (CHU Henri Mondor), Odile Goria (Rouen), Gilbert Habib (Marseille, La Timone), Pauline Houssel-Debry (Rennes), Etienne-Marie Jutant (Kremlin Bicêtre), Sylvie Leroy (Nice), Pierre Mauran (Reims), Jean-François Mornex (Lyon), Patrice Poubeau (La Réunion), François Picard (Bordeaux), Christophe Pison (Grenoble), Martine Reynaud-Gaubert (Marseille, Nord), Florence Parent (Kremlin Bicêtre), Pascal Roblot (Poitiers), Andrei Seferian (Kremlin Bicêtre)

#### References

- 1. Savale L, Watherald J, Sitbon O. Portopulmonary Hypertension. Semin Respir Crit Care Med. 2017;38(5):651–61.
- 2. Asrani SK, Devarbhavi H, Eaton J, Kamath PS. Burden of liver diseases in the world. J Hepatol. 2019;70(1):151–71.
- 3. Le Pavec J, Souza R, Herve P, Lebrec D, Savale L, Tcherakian C, et al. Portopulmonary hypertension: survival and prognostic factors. Am J Respir Crit Care Med. 2008;178(6):637–43.
- 4. Krowka MJ, Miller DP, Barst RJ, Taichman D, Dweik RA, Badesch DB, et al. Portopulmonary hypertension: a report from the US-based REVEAL Registry. Chest. 2012;141(4):906–15.
- Sithamparanathan S, Nair A, Thirugnanasothy L, Coghlan JG, Condliffe R, Dimopoulos K, et al. Survival in portopulmonary hypertension: Outcomes of the United Kingdom National Pulmonary Arterial Hypertension Registry. J Heart Lung Transplant. 2017;36(7):770–9.
- Krowka MJ, Fallon MB, Kawut SM, Fuhrmann V, Heimbach JK, Ramsay MAE, et al. International Liver Transplant Society Practice Guidelines: Diagnosis and Management of Hepatopulmonary Syndrome and Portopulmonary Hypertension. Transplantation. 2016;100(7):1440–52.
- 7. Galiè N, Humbert M, Vachiery J-L, Gibbs S, Lang I, Torbicki A, et al. 2015 ESC/ERS Guidelines for the diagnosis and treatment of pulmonary hypertension: The Joint Task Force for the Diagnosis and Treatment of Pulmonary Hypertension of the European Society of Cardiology (ESC) and the European Respiratory Society (ERS): Endorsed by: Association for European Paediatric and Congenital Cardiology (AEPC), International Society for Heart and Lung Transplantation (ISHLT). Eur Respir J. 2015;46(4):903–75.
- 8. Galiè N, Humbert M, Vachiery J-L, Gibbs S, Lang I, Torbicki A, et al. 2015 ESC/ERS Guidelines for the diagnosis and treatment of pulmonary hypertension: The Joint Task Force for the Diagnosis and Treatment of Pulmonary Hypertension of the European Society of Cardiology (ESC) and the European Respiratory Society (ERS): Endorsed by: Association for European Paediatric and Congenital Cardiology (AEPC), International Society for Heart and Lung Transplantation (ISHLT). Eur Heart J. 2016;37(1):67–119.
- 9. Sitbon O, Bosch J, Cottreel E, Csonka D, de Groote P, Hoeper MM, et al. Macitentan for the treatment of portopulmonary hypertension (PORTICO): a multicentre, randomised, double-blind, placebo-controlled, phase 4 trial. Lancet Respir Med. 2019;7(7):594–604.
- Hoeper MM, Kramer T, Pan Z, Eichstaedt CA, Spiesshoefer J, Benjamin N, et al. Mortality in pulmonary arterial hypertension: prediction by the 2015 European pulmonary hypertension guidelines risk stratification model. Eur Respir J. 2017;50(2):1700740.

- 11. Escribano-Subias P, Blanco I, López-Meseguer M, Lopez-Guarch CJ, Roman A, Morales P, et al. Survival in pulmonary hypertension in Spain: insights from the Spanish registry. Eur Respir J. 2012;40(3):596–603.
- 12. Ling Y, Johnson MK, Kiely DG, Condliffe R, Elliot CA, Gibbs JSR, et al. Changing demographics, epidemiology, and survival of incident pulmonary arterial hypertension: results from the pulmonary hypertension registry of the United Kingdom and Ireland. Am J Respir Crit Care Med. 2012;186(8):790–6.
- 13. Galiè N, Channick RN, Frantz RP, Grünig E, Jing ZC, Moiseeva O, et al. Risk stratification and medical therapy of pulmonary arterial hypertension. Eur Respir J. 2019;53(1):1801889.
- 14. Savale L, Magnier R, Le Pavec J, Jaïs X, Montani D, O'Callaghan DS, et al. Efficacy, safety and pharmacokinetics of bosentan in portopulmonary hypertension. Eur Respir J. 2013;41(1):96–103.
- Sitbon O, Sattler C, Bertoletti L, Savale L, Cottin V, Jaïs X, et al. Initial dual oral combination therapy in pulmonary arterial hypertension. Eur Respir J. 2016;47(6):1727– 36.
- 16. Touma W, Nayak RP, Hussain Z, Bacon BR, Kudva GC. Epoprostenol-induced hypersplenism in portopulmonary hypertension. Am J Med Sci. 2012;344(5):345–9.
- 17. Wang Y-W, Lin H-C, Yang Y-Y, Hou M-C, Lee S-D. Sildenafil decreased pulmonary arterial pressure but may have exacerbated portal hypertension in a patient with cirrhosis and portopulmonary hypertension. J Gastroenterol. 2006;41(6):593–7.
- 18. Kalambokis GN, Kosta P, Pappas K, Tsianos EV. Haemodynamic and renal effects of tadalafil in patients with cirrhosis. World J Gastroenterol. 2010;16(39):5009–10.
- 19. Krowka MJ, Plevak DJ, Findlay JY, Rosen CB, Wiesner RH, Krom RA. Pulmonary hemodynamics and perioperative cardiopulmonary-related mortality in patients with portopulmonary hypertension undergoing liver transplantation. Liver Transplant. 2000;6(4):443–50.
- 20. DeMartino ES, Cartin-Ceba R, Findlay JY, Heimbach JK, Krowka MJ. Frequency and Outcomes of Patients With Increased Mean Pulmonary Artery Pressure at the Time of Liver Transplantation. Transplantation. 2017;101(1):101–6.
- 21. Khaderi S, Khan R, Safdar Z, Stribling R, Vierling JM, Goss JA, et al. Long-term followup of portopulmonary hypertension patients after liver transplantation. Liver Transplant. 2014;20(6):724–7.
- 22. Savale L, Sattler C, Coilly A, Conti F, Renard S, Francoz C, et al. Long-term outcome in liver transplantation candidates with portopulmonary hypertension. Hepatol. 2017;65(5):1683–92.
- 23. Swanson KL, Wiesner RH, Nyberg SL, Rosen CB, Krowka MJ. Survival in portopulmonary hypertension: Mayo Clinic experience categorized by treatment

subgroups. Am J Transplant. 2008;8(11):2445-53.

- 24. Benza RL, Gomberg-Maitland M, Elliott CG, Farber HW, Foreman AJ, Frost AE, et al. Predicting Survival in Patients With Pulmonary Arterial Hypertension: The REVEAL Risk Score Calculator 2.0 and Comparison With ESC/ERS-Based Risk Assessment Strategies. Chest. 2019;156(2):323-337.
- 25. Boucly A, Weatherald J, Savale L, Jaïs X, Cottin V, Prevot G, et al. Risk assessment, prognosis and guideline implementation in pulmonary arterial hypertension. Eur Respir J. 2017;50(2):1700889.
- 26. Weatherald J, Boucly A, Launay D, Cottin V, Prévot G, Bourlier D, et al. Haemodynamics and serial risk assessment in systemic sclerosis associated pulmonary arterial hypertension. Eur Respir J. 2018;52(4):1800678.
- 27. Weatherald J, Boucly A, Chemla D, Savale L, Peng M, Jevnikar M, et al. Prognostic Value of Follow-Up Hemodynamic Variables After Initial Management in Pulmonary Arterial Hypertension. Circulation. 2018;137(7):693–704.

Variables	
Age, years	55±10
Male sex, $n(\%)$	372 (58)
Body Mass Index, $kg.m^{-2}(n=617)$	27.2±6
Etiology of portal hypertension, n (%) (n=637)	
Cirrhosis	622 (97.6)
Alcohol	370 (58.1)
Viral hepatitis	83 (13)
Mixed viral and alcohol	100 (15.7)
Autoimmune	21 (3.3)
Dysmetabolic	22 (3.5)
Cryptogenic	15 (2.3)
Others	11 (1.7)
Extrahepatic portal hypertension	10 (1.6)
Congenital portosystemic shunt	5 (0.8)
Severity of cirrhosis	
Child-Pugh stage, $n$ (%) ( $n=577$ )	
Stage A	328 (56.8)
Stage B	192 (33.3)
Stage C	57 (9.9)
MELD score $(n=495)$	
Median (Q1, Q3)	11 (9, 15)
MELD score $\geq$ 15, <i>n</i> (%)	148 (30)
Hepatocarcinoma, n (%)	76 (11.9)
HIV infection, <i>n</i> (%)	49 (7.7)
Indication for liver transplantation, <i>n</i> (%)	94 (14.7)
NYHA functional class, $n$ (%) ( $n=636$ )	
Ι	34 (5.3)
П	199 (31.2)
III	342 (53.8)
IV	61 (9.6)
6-min walk distance, m ( $n=509$ )	355 (252, 425)
BNP <50 ng.L <sup>-1</sup> or NT-proBNP <300 ng.L <sup>-1</sup> , <i>n</i> (%) ( <i>n</i> =498)	112 (22.5)
Hemodynamics variables	
Right atrial pressure, $mmHg$ ( $n=610$ )	7 (5-11)
Mean pulmonary artery pressure, <i>mmHg</i>	47±11
Pulmonary artery wedge pressure, mmHg	9±3
Cardiac output, L.min <sup>-1</sup>	5.3±1.6
Cardiac index, L.min <sup>-1</sup> per $m^2$ (n=624)	2.8±0.8
Mixed venous oxygen saturation, $\%$ ( <i>n</i> =332)	64±9
Dulmonory vescular registence. Wood Units	7 (5 2 10)

#### **Table 1. Baseline characteristics**

Pulmonary vascular resistance, Wood Units7 (5.2-10)Variables are expressed as median (interquartile range) or mean±SD as appropriate. Number<br/>of patients is 637 unless otherwise stated. Abbreviations: BNP, Brain Natriuretic Peptide;<br/>HIV, human immunodeficiency virus; MELD, model for end-stage liver disease; NYHA,<br/>New York Heart Association; NT-proBNP, N-terminal pro-Brain Natriuretic Peptide.

	n	Baseline	First follow-	P-value
		visit	up visit	
NYHA functional class, n (%)	504			
Ι		24 (5)	73 (14)	
П		154 (31)	313 (63)	<0.0001
III		280 (55)	106 (21)	
IV		46 (9)	12 (2)	
6-minute walk distance, <i>m</i>	349	360	403	<0.0001
		(270-430)	(329-462)	
BNP <50 ng.L <sup>-1</sup> or NT-proBNP <300 ng.L <sup>-1</sup> ,	317	72 (23)	143 (45)	<0.0001
n (%)				
Hemodynamic variables	453			
Right atrial pressure, mmHg		7 (5-11)	6 (4-9)	<0.0001
Mean pulmonary artery pressure, mmHg		48±11	39±11	<0.0001
Pulmonary artery wedge pressure, mmHg		9±3	10 <b>±</b> 4	<0.0001
Cardiac output, L.min <sup>-1</sup>		5.3±1.7	6.9±2.1	<0.0001
Cardiac index, $L.min^{-1} per m^2$		2.8±0.8	3.7±1	<0.0001
Pulmonary vascular resistance, Wood Units		7.1 (5.3-10.2)	4.1 (2.8-5.7)	<0.0001
Mixed venous oxygen saturation, %		64±9	71±7	<0.0001

Variables are expressed as mean  $\pm$  SD or median (interquartile range) as appropriate. Comparisons made using a paired Student's t test, Wilcoxon signed-rank test, or Chisquare test as appropriate.

*Abbreviations*: BNP, Brain Natriuretic Peptide; NT-proBNP; N-terminal pro-brain natriuretic peptide; NYHA, New York Heart Association; PAH, Pulmonary arterial hypertension.

	Univariate analysis		Multivariate analysis (model A)		Multivariate analysis (model B)	
	Hazard ratio (95%CI)	p-value	Hazard ratio (95%CI)	p-value	Hazard ratio (95%CI)	p-value
Age (year)	1.022 (1.011-1.034)	0.0001	1.022 (1.002-1.041)	0.0302	1.022 (1.004-1.04)	0.0158
Female sex	0.801 (0.635-1.009)	0.0599	0.493 (0.332-0.733)	0.0005	0.616 (0.436-0.871)	0.0061
Body Mass Index (kg.m <sup>-2</sup> )	0.995 (0.975-1.015)	0.6219				
NYHA-FC III-IV vs I-II	1.342 (1.055-1.716)	0.0168	1.164 (0.781-1.736)	0.4555	1.368 (0.958-1.955)	0.8480
6-minute walk distance (m)	0.998 (0.997-0.998)	<0.0001	0.997 (0.996-0.999)	0.0002	0.998 (0.997-0.999)	0.0007
Right atrial pressure (mmHg)	1.024 (1.007-1.042)	0.0063	1.005 (0.978-1.033)	0.7267	1.001 (0.977-1.026)	0.9233
mPAP (mmHg)	0.996 (0.986-1.006)	0.4097				
PAWP (mmHg)	0.988 (0.953-1.024)	0.5030				
Cardiac output (L.min <sup>-1</sup> )	1.004 (0.936-1.078)	0.9047				
Cardiac index (L.min <sup>-1</sup> per m <sup>2</sup> )	1.048 (0.911-1.205)	0.5150				
PVR (Wood unit)	0.990 (0.965-1.016)	0.4554				
SvO <sub>2</sub> (%)	0.994 (0.977-1.011)	0.4589				
BNP <50 or NT-proBNP <300 ng/L	0.712 (0.512-0.99)	0.0433	0.956 (0.606-1.508)	0.8480	0.948 (0.630-1.426)	0.7975
Creatinine (µg/L)	1.003 (1.001-1.006)	0.0068				
Total bilirubin (µg/L)	1.006 (1.004-1.008)	<0.0001				
Extrahepatic PoH or congenital	0.351 (0.113-1.096)	0.0716				
portosystemic shunt						
MELD score (per unit)	1.081 (1.056-1.106)	<0.0001	1.062 (1.026-1.098)	0.0005		
Child-Pugh stage B/C vs A	2.451) (1.930-3.112)	<0.0001			1.924 (1.385-2.672)	<0.0001
Hepatocarcinoma, yes	1.429 (1.022-1.996)	0.0368	0.956 (0.568-1.608)	0.8640	0.965 (0.597-1.559)	0.8847
HIV infection	1.019 (0.654-1.589)	0.9330				
Initiation of PAH-targeted therapy, no	1.517 (1.083-2.126)	0.0153	1.503 (0.76-2.973)	0.2412	1.65 (0.965-2.823)	0.0674

Table 3. Univariate and multivariate analyses of baseline variables for overall survival

*Abbreviations*: BNP: Brain Natriuretic Peptide; 95% CI: 95% confidence interval; HIV: Human immunodeficiency virus; HR: Hazard Ratio; MELD: Model for end-stage liver disease; mPAP: Mean pulmonary artery pressure; NT-proBNP: N-terminal proBNP; NYHA-FC: New York Heart Association Functional Class; PAWP: pulmonary artery wedge pressure; PoH: portal hypertension; PVR: pulmonary vascular resistance; SvO<sub>2</sub>: Mixed venous oxygen saturation.

#### **Figure Legends**

#### **Figure 1. Patient disposition**

*Abbreviations*: ERA, endothelin receptor antagonist; PAH, pulmonary arterial hypertension; PAWP, pulmonary artery wedge pressure; PDE-5i, phosphodiesterase type-5 inhibitor; IV PCA, intravenous prostacyclin; PH, pulmonary hypertension; PoPH, portopulmonary hypertension; PVR, pulmonary vascular resistance; RHC: right heart catheterization; WU, Wood unit.

# Figure 2. Short-term effect of initial treatment strategy on cardiopulmonary haemodynamics and 6-minute walk distance

Changes from baseline (T0) to first follow-up (T1) were analysed using the paired Student's ttest or Wilcoxson signed-rank test, as appropriate: \*\*\* p<0.001; \*\*\*\* p<0.0001 Baseline values were compared using the unpaired Student's t-test or Mann-Whitney U test as appropriate: NS, non-significant; † p<0.05; †† p<0.01; ††† p<0.001; †††† p<0.0001 Changes in hemodynamic variables and 6-min walk distance were compared using the Mann-Whitney test: no significant difference between ERA and PDE-5i in monotherapy; significant differences between combination therapy and monotherapy (ERA or PDE-5i) (‡‡ p<0.01,  $\ddagger\ddagger p<0.0001$ ).

*Abbreviations*: 6MWD, 6-minute walk distance; CO, cardiac output; ERA, endothelin receptor antagonist; mPAP, mean pulmonary arterial pressure; PDE-5i, phosphodiesterase type-5 inhibitor; PVR, pulmonary vascular resistance

Figure 3. Outcomes of PoPH cohort. (A) Overall survival of the PoPH cohort; (B) Survival of patients with cirrhosis and of those with extrahepatic portal hypertension or congenital portosystemic shunt; (C) Survival according to Child-Pugh stage at the time of PoPH diagnosis; (D) Survival according to MELD score at the time of PoPH diagnosis.

*Abbreviations*: CP, Child-Pugh; CPS, congenital portosystemic shunt; MELD, Model for End-Stage Liver Disease; PoH, portal hypertension

Figure 4. Outcomes of liver transplant candidates (A) Survival of PoPH patients by liver transplant status, (B) Survival of PoPH patients from liver transplantation

Figure 5. Survival of transplanted and non-transplanted patients according to Child-Pugh stage and MELD score

















В





---- Non-transplanted

— Transplanted

Patients at risk,

Non-transplanted	211	143	105	79	55	34
— Transplanted	38	34	26	20	19	12

Cohort of 637 patients with portopulmonary hypertension

*Initial PAH therapy in 574 patients* (90.1%)



# Endothelin receptor antagonist







